Author's response to reviews

**Title:** Removal of a below knee plaster cast worn for twenty eight months: a case report

**Authors:**

Helen Ingoe (Helen_ingoe@hotmail.com)
Sarah Eastwood (sarah.eastwood@talk21.com)
David W Elson (davidelson@yahoo.com)
Claire F Young (claire.young@ncuh.nhs.uk)

**Version:** 3 **Date:** 11 November 2010

**Author's response to reviews:**

The following changes to formatting have been made

The middle initials of the 3rd and 4th author have been included throughout and updated on the author list.

The abstract has the correct structure: Introduction, Case Presentation, Conclusion.

Details of the patient’s ethnicity, age and gender have been added to both abstract and case presentation sections.

The country (united kingdom) added to affiliation details

All text now in black